A randomised, double blind, placebo-controlled trial of a two-week course of dexamethasone for adult patients with a symptomatic Chronic Subdural Haematoma (Dex-CSDH trial)

Peter J Hutchinson,^{1*} Ellie Edlmann,^{1,2,3} John G Hanrahan,¹ Diederik Bulters,⁴ Ardalan Zolnourian,⁴ Patrick Holton,⁴ Nigel Suttner,⁵ Kevin Agyemang,⁵ Simon Thomson,⁶ Ian A Anderson,⁶ Yahia Al-Tamimi,⁷ Duncan Henderson,⁷ Peter Whitfield,³ Monica Gherle,³ Paul M Brennan,⁸ Annabel Allison,⁹ Eric P Thelin,^{1,10} Silvia Tarantino,¹ Beatrice Pantaleo,⁹ Karen Caldwell,¹ Carol Davis-Wilkie,9 Harry Mee,1 Elizabeth A Warburton,¹ Garry Barton,¹¹ Aswin Chari,¹² Hani J Marcus,¹³ Sarah Pyne,¹¹ Andrew T King,¹⁴ Antonio Belli,¹⁵ Phyo K Myint,¹⁶ Ian Wilkinson,^{9,17} Thomas Santarius,¹ Carole Turner,¹ Simon Bond⁹ and Angelos G Kolias¹ on behalf of the British Neurosurgical Trainee Research Collaborative and Dex-CSDH Trial Collaborators[†]

¹Division of Neurosurgery, Department of Clinical Neurosciences, Addenbrooke's Hospital & University of Cambridge, Cambridge Biomedical Campus, Cambridge, UK ²Faculty of Health, Medicine, Dentistry and Human Sciences, University of Plymouth, Plymouth, UK

- ³South West Neurosurgical Centre, Derriford Hospital, Plymouth, UK
- ⁴Wessex Neurological Unit, University Hospital Southampton, Southampton, UK
- ⁵Department of Neurosurgery, Queen Elizabeth University Hospital, Glasgow, UK
- ⁶Department of Neurosurgery, Leeds General Infirmary, Leeds, UK
- ⁷Department of Neurosurgery, Royal Hallamshire Hospital, Sheffield, UK
- ⁸Translational Neurosurgery, Centre for Clinical Brain Sciences, University of Edinburgh, Edinburgh, UK
- ⁹Cambridge Clinical Trials Unit, Cambridge University Hospitals NHS Foundation Trust, Cambridge, UK
- ¹⁰Department of Neurology, Karolinska University Hospital, Stockholm, Sweden

- ¹¹Norwich Medical School, Faculty of Medicine and Health Sciences, University of East Anglia, Norfolk, UK
- ¹²Department of Neurosurgery, Great Ormond Street Hospital & Institute of Child Health, University College London, London, UK
- ¹³National Hospital for Neurology and Neurosurgery, London, UK
- ¹⁴Department of Neurosurgery, Manchester Centre for Clinical Neurosciences, Salford Royal Hospital, Manchester Academic Health Science Centre, Manchester, UK
- ¹⁵NIHR Surgical Reconstruction and Microbiology Research Centre and Institute of Inflammation and Ageing, University of Birmingham, Birmingham, UK
- ¹⁶Ageing Clinical & Experimental Research Group, Institute of Applied Health Science, University of Aberdeen, Aberdeen, UK
- ¹⁷Division of Experimental Medicine and Immunotherapeutics, Addenbrooke's Hospital, University of Cambridge, Cambridge, UK

*Corresponding author

[†]Listed in Appendix 1

Disclosure of interests

Full disclosure of interests: Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/XWZN4832.

Primary conflicts of interest: Peter J Hutchinson was a member of the Health Technology Assessment (HTA) Programme Intellectual Property Panel (January 2016 to May 2018), and is currently a member of the HTA Prioritisation Committee B (in hospital). Antonio Belli reports being employed by and owning shares in Marker Diagnostics (Birmingham, UK). Elizabeth A Warburton is currently a member of the HTA Clinical Evaluation and Trials Committee (2019–present). Garry Barton was a member of the Norwich Clinical Trials Unit funded by the National Institute for Health and Care Research (NIHR) (until August 2021). Ian Wilkinson was a member of the Efficacy and Mechanism Evaluation Funding Committee (May 2015 to May 2019). Ellie Edlmann reports grants from Royal College of Surgeons and Rosetrees Trust Research Fellowship, during the conduct of the study. Carol Davis-Wilkie was funded as an employee of the Cambridge Clinical Trials Unit.

Published March 2024 DOI: 10.3310/XWZN4832

Plain language summary

A randomised, double blind, placebo-controlled trial of a two-week course of dexamethasone for adult patients with a symptomatic Chronic Subdural Haematoma (Dex-CSDH trial)

Health Technology Assessment 2024; Vol. 28: No. 12 DOI: 10.3310/XWZN4832

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain language summary

Chronic subdural haematoma is one of the most common conditions managed in adult neurosurgery and mainly affects older people. It is an 'old' collection of blood and blood breakdown products found on the surface of the brain. Surgery to drain the liquid collection is effective, with most patients improving. Given that inflammation is involved in the disease process, a commonly used steroid, dexamethasone, has been used alongside surgery or instead of surgery since the 1970s. However, there is no consensus or high-quality studies confirming the effectiveness of dexamethasone for the treatment of chronic subdural haematoma.

This study was designed to determine the effectiveness of adding dexamethasone to the normal treatment for patients with a symptomatic chronic subdural haematoma. The benefit of adding dexamethasone was measured using a disability score called the Modified Rankin Scale, which can be divided into favourable and unfavourable outcomes. This was assessed at 6 months after entry into the study.

In total, 748 adults with a symptomatic chronic subdural haematoma treated in neurosurgical units in the UK participated. Each participant had an equal chance of receiving either dexamethasone or a placebo because they were assigned randomly. Neither the patients nor the investigators knew who received dexamethasone and who received placebo.

Most patients in both groups had an operation to drain the haematoma and experienced significant functional improvement at 6 months compared with their initial admission to hospital. However, patients who received dexamethasone had a lower chance than patients who received placebo of favourable recovery at 6 months. Specifically, 84% of patients who received dexamethasone had recovered well at 6 months, compared with 90% of patients who received placebo. There were more complications in the group that received dexamethasone.

This trial demonstrates that adding dexamethasone to standard treatment reduced the chance of a favourable outcome compared with standard treatment alone. Therefore, this study does not support the use of dexamethasone in treating patients with a symptomatic chronic subdural haematoma.

Health Technology Assessment

ISSN 2046-4924 (Online)

Impact factor: 3.6

A list of Journals Library editors can be found on the NIHR Journals Library website

Launched in 1997, *Health Technology Assessment* (HTA) has an impact factor of 3.6 and is ranked 32nd (out of 105 titles) in the 'Health Care Sciences & Services' category of the Clarivate 2022 Journal Citation Reports (Science Edition). It is also indexed by MEDLINE, CINAHL (EBSCO Information Services, Ipswich, MA, USA), Embase (Elsevier, Amsterdam, the Netherlands), NCBI Bookshelf, DOAJ, Europe PMC, the Cochrane Library (John Wiley & Sons, Inc., Hoboken, NJ, USA), INAHTA, the British Nursing Index (ProQuest LLC, Ann Arbor, MI, USA), Ulrichsweb™ (ProQuest LLC, Ann Arbor, MI, USA) and the Science Citation Index Expanded™ (Clarivate™, Philadelphia, PA, USA).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta.

Criteria for inclusion in the Health Technology Assessment journal

Manuscripts are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This manuscript

The research reported in this issue of the journal was funded by the HTA programme as award number 13/15/02. The contractual start date was in September 2014. The draft report began editorial review in January 2021 and was accepted for publication in March 2022. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' manuscript and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this manuscript.

This manuscript presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the NHS, those of the authors, those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care.

Copyright © 2024 Hutchinson *et al.* This work was produced by Hutchinson *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland, and final files produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).